Author's response to reviews

Title: Technical development of PubMed Interact: an improved interface for MEDLINE/PubMed searches

Authors:

Michael Muin (mmuin@mail.nih.gov)
Paul Fontelo (fontelo@nlm.nih.gov)

Version: 4 Date: 2 November 2006

Author's response to reviews: see over
Dear BMC Editors,

We are resubmitting our manuscript entitled, ‘Technical development of PubMed Interact: an improved interface for MEDLINE/PubMed searches’.

All formatting changes requested by the editorial production team have been carried out and addressed.

Thank you.

With great respect,

Michael Muin
Paul Fontelo
Formatting changes requested
----------------------------
Website addresses - Websites must be removed from the text, and cited from the main reference list only. If you are using referencing software such as EndNote or Refman, please ensure that your references are entered through the software rather than included in the manuscript reference list manually. Please present your reference in the following structure, e.g.
17. Website title [www.website.com].
Where possible, please avoid using long web addresses. (NB: There are two exceptions to this rule:
When the abstract contains a weblink.
When a URL is included in an Availability and Requirements section).

Response: Done. URL hyperlinks in the Acknowledgement were taken out and put in References. A URL on page 13 was taken out.

Authors' contributions - Please indicate that all authors read and approved the final manuscript.

Response: Done. Added one sentence in the paragraph.

Reference 1 - Please put the author surnames before the first initial.

Response: Done.